Author’s response to reviews

Title: Expert consensus on relevant risk predictors for the occurrence of osteoporotic fractures in specific clinical subgroups – Delphi Survey

Authors:

Nicolas Bodmer (bodmer@medignition.ch)

HansJörg Häuselmann (hjhauselmann@rheumazentrum.ch)

Diana Frey (diana.frey@usz.ch)

Daniel Aeberli (daniel.aeberli@insel.ch)

Lucas Bachmann (bachmann@medignition.ch)

Version: 2 Date: 22 Aug 2019

Author’s response to reviews:

Dear Dr. Stein, dear members of the Editorial Board,

Thank you very much for considering our paper as potentially acceptable for publication and for your contributions to improve the manuscript. Please find attached a point-by-point reply of the comments and a revised clean version of the manuscript.

We all look forward to hearing from you again soon.

Also on behalf of my co-authors,

Yours sincerely,

Nicolas S. Bodmer

Point-to-Point-reply

Editor Comments: Most comments have been addressed. The authors should add a sentence to the methods explaining that the panel was told they would be anonymous and why this decision was made.

Our Reply:

Method section, line 86: Anonymous participation was granted to ensure freedom of expression.